DISCUSSION PAPER SERIES

No. 2711

MODELLING SCALE CONSISTENT VAR WITH THE TRUNCATED LÉVY FLIGHT

Thorsten Lehnert and Christian C P Wolff

FINANCIAL ECONOMICS





www.cepr.org

Available online at:

www.cepr.org/pubs/dps/DP2711.asp

MODELLING SCALE CONSISTENT VAR WITH THE TRUNCATED LÉVY FLIGHT

Thorsten Lehnert, Limburg Institute of Financial Economics, Maastricht University Christian C P Wolff, Limburg Institute of Financial Economics, Maastricht University and CEPR

> Discussion Paper No. 2711 February 2001

Centre for Economic Policy Research 90–98 Goswell Rd, London EC1V 7RR, UK Tel: (44 20) 7878 2900, Fax: (44 20) 7878 2999 Email: cepr@cepr.org, Website: www.cepr.org

This Discussion Paper is issued under the auspices of the Centre's research programme in **Financial Economics**. Any opinions expressed here are those of the author(s) and not those of the Centre for Economic Policy Research. Research disseminated by CEPR may include views on policy, but the Centre itself takes no institutional policy positions.

The Centre for Economic Policy Research was established in 1983 as a private educational charity, to promote independent analysis and public discussion of open economies and the relations among them. It is pluralist and non-partisan, bringing economic research to bear on the analysis of medium- and long-run policy questions. Institutional (core) finance for the Centre has been provided through major grants from the Economic and Social Research Council, under which an ESRC Resource Centre operates within CEPR; the Esmée Fairbairn Charitable Trust; and the Bank of England. These organizations do not give prior review to the Centre's publications, nor do they necessarily endorse the views expressed therein.

These Discussion Papers often represent preliminary or incomplete work, circulated to encourage discussion and comment. Citation and use of such a Paper should take account of its provisional character.

Copyright: Thorsten Lehnert and Christian C P Wolff

CEPR Discussion Paper No. 2711

February 2001

ABSTRACT

Modelling Scale Consistent VaR with the Truncated Lévy Flight*

Returns in financial assets show consistent excess kurtosis, indicating the presence of large fluctuations not predicted by Gaussian models. Mandelbrot (1963) first proposed the idea that price changes distributed according to a Lévy stable law. The unique feature of Lévy-stable distributions in general is that they are stable under addition. However, these distributions have power law tails that decay too slowly from the point of view of financial modelling. In recent studies the truncated Lévy flight has been shown to eliminate this problem and to be very promising for the modelling of financial dynamics. An exponential decay in the tails ensures that all moments are finite and the distribution is fat-tailed for short time scales and converges in a Gaussian process for increasing time scales, a feature observed in financial data. We propose a model with time-varying scale parameter (GARCH process) error terms that are truncated Lévy distributed. We determine the appropriate GARCH specification for each data set by conducting a specification test based on a generalization of the augmented GARCH process of Duan (1997). The Lévy flight includes a method for scaling up a single-day volatility to a multi-day volatility, precisely a α -root-of-time rule, where α is the characteristic parameter of the process. We use this rule to forecast future volatility and as a result estimate value-at-risk (VaR) several days ahead and compare it to the RiskMetricsTM (1996) approach, which is a special case of our method. We compare the models in in-sample and out-of-sample analyses for a sample of stock index returns.

JEL Classification: C22, C52, G10

Keywords: augmented GARCH process, in- and out-of-sample analysis, scale consistency, truncated Lévy flight, value-at-risk

Thorsten Lehnert Limburg Institute of Financial Economics Maastricht University PO Box 616 6200 MD Maastricht THE NETHERLANDS Tel: (31 43) 388 3859 Fax: (31 43) 388 4875 Email: t.lehnert@berfin.unimaas.nl

Christian C P Wolff Limburg Institute of Financial Economics Maastricht University PO Box 616 6200 MD Maastricht THE NETHERLANDS Tel: (31 43) 388 3838 Fax: (31 43) 388 4875 Email: c.wolff@berfin.unimaas.nl

For further Discussion Papers by this author see: www.cepr.org/pubs/new-dps/dplist.asp?authorid=105187

*We would like to thank the participants at the 2000 World Congress of the Bachelier Finance Society in Paris for helpful comments and suggestions.

Submitted 21 December 2000

NON-TECHNICAL SUMMARY

In this Paper we consider the price dynamics of stock indices traded in a financial market. While financial applications involve many different time intervals, ranging from a few minutes (intraday) to a number of years, most techniques used in econometrics focus on modelling the fluctuations of price series in a single time interval. But the distribution that successfully explains daily price changes, for example, is unable to characterize the nature of hourly price changes. On the other hand, the statistical properties of monthly price changes are often not fully covered by a model for daily price changes. In order to describe the statistics of future prices of a financial asset, a distribution of price fluctuations for different time intervals is *a priori* needed, corresponding to different trading time horizons.

For derivative pricing and risk management a better description of the extreme events is crucial. There is a need for an alternative to the Gaussian with more weight in the tails and, just as importantly, more weight around the centre. A possible candidate is the Lévy distribution often studied in physics. Fitting the Lévy distribution to the data seems to provide a consistently better representation of price fluctuations. In practice it is possible to model the tails as fat as necessary by modifying only one parameter (the Gaussian distribution with thin tails is just a special case).

The problems with these kinds of distributions are the power law tails, which decay too slowly from the point of view of financial modelling and the fact that the distribution of price changes for larger time intervals converge in a Gaussian distribution model. What is observed on financial markets is a faster decay like the exponential decay of the Gaussian model. This problem can be overcome simply by taking the Lévy distribution in the central part and introducing a cut-off in the far tails that is faster than the Lévy power law tails. The Lévy distribution with a cut-off and exponentially declining tails was introduced by Mantegna and Stanley (1994) and is known as a truncated Lévy distribution (TLD). Koponen (1995) derived an analytical expression of the characteristic function. This cut-off ensures that the variance will be finite and the distribution converges to a Gaussian one in the end. To model financial prices over time the truncated Lévy flight (TLF) can be constructed by the sum of independent and identically distributed random variables described by a TLD. In contrast to fat-tailed distributions, the central limit theorem applies and the TLF approaches a Gaussian model. Lévy flights have been observed experimentally in physical systems and have been used to describe for instance the spectral random walk of a single molecule embedded in a solid. In these cases an unavoidable cut-off is always present.

In contrast to physical systems, hundreds of papers have shown that there is a strong non-IID clustering effect in financial data. Most of the studies so far performed a summability test of correlated unconditional return data and rejected the Lévy-stable hypothesis. But testing for stable properties requires IID data. For example, Ghose and Kroner (1995) found that summability holds for data generated from an IGARCH process with non-normal innovations. But of course the simulated data is correlated and not IID. Therefore if we want to justify the use of Lévy processes we first have to remove the autocorrelation in the data and second, assume the summability of the IID data. A solution to this problem would be a mechanism to 'filter out' the correlations in the data and test if the IID residual process is adequately described by the stable law.

Fortunately, the class of GARCH models has been very successful in modelling the significant volatility clustering and non-IID nature of the data. More specifically, the standard GARCH model produces a mean-reverting time-dependent volatility process that 'filters out' the correlations in the data and the remaining residuals are assumed to be IID. We develop a specification test based on the augmented GARCH process of Duan (1997) to derive the appropriate model for each data set and each distribution. We show that different choices for the innovations distribution lead to different results about the ability of the GARCH model to explain the autocorrelations in the data. In particular, we show that asymmetry in the volatility is not enough to capture the skewness in the data and a combination of a skewed distribution for the innovations and asymmetry in the volatility process leads to a much better fit in-sample and to better out-of-sample performance. In general, if the model residuals can be shown to be IID, we can assume the summability of the residuals. If we can assume that the residuals have stable properties we can use the alpha-root-of-time rule for scaling up single-day volatility to multiday volatility.

A practical risk management tool for financial institutions nowadays is valueat-risk (VaR). It is prescribed by the Basle Committee to report to the (international) supervisor and also used as an internal management tool to see if traders are within their limits. VaR is defined as an amount lost on a portfolio with a given probability over a fixed number of days. The confidence level reflects 'extreme market conditions' with a probability of, for example, 99%. In other words in only 1% of the cases we will lose more than the reported VaR of our portfolio. Crucial for an accurate VaR estimate is the precise determination of the 'extreme events', which results from the distribution function of the underlying return series in the tails. Since the Lévy distribution overestimates and the Gaussian underestimates the fat tails, we expect the same behaviour for the true value-at-risk at high quantiles. To capture the real downside risk, a parametric distribution with fatter tails than the Gaussian and thinner tails than the Lévy should be used. It should be possible to fit the tails separately just to the extreme events. We show that especially for the maximum likelihood method it is crucial to use a skewed distribution to capture the downside risk. For this reason the truncated Lévy distribution is an attractive statistical tool for estimating value-at-risk. We use the truncated Lévy flight as a stochastic process for different time intervals and model the time-scaling behaviour of the quantiles.

1 INTRODUCTION

In this paper we consider the price dynamics of stock indices traded in a financial market. While financial applications involve many different time intervals, ranging from a few minutes (intraday) to a number of years, most techniques used in econometrics focus on modeling the fluctuations of price series in a single time interval. But the distribution that successfully explains daily price changes, for example, is unable to characterize the nature of hourly price changes. On the other hand, the statistical properties of monthly price changes are often not fully covered by a model for daily price changes. In order to describe the statistics of future prices of a financial asset, one a priori needs a distribution of price fluctuations for different time intervals, corresponding to different trading time horizons.

For derivative pricing and risk management a better description of the large events is crucial. There is a need of an alternative to the Gaussian with more weight in the tails and also important more weight around the center. A possible candidate is the Lévy distribution often studied in physics. Fitting the Lévy distribution to the data seems to provide a consistently better representation of price fluctuations. In practice it is possible to model the tails as fat as necessary by modifying only one parameter (the Gaussian distribution with thin tails is just a special case).

The problems with these kinds of distributions are the power law tails, which decay too slowly from the point of view of financial modeling and the fact that the distribution of price changes for larger time intervals converge to a Gaussian. What is observed on financial markets is a faster decay like the exponential decay of the Gaussian. This problem can be simply overcome by taking the Lévy distribution in the central part and introducing a cutoff in the far tails that is faster than the Lévy power law tails. The Lévy distribution with a cutoff and exponentially declining tails was introduced by Mantegna/Stanley (1994) and is known as a truncated Lévy distribution (TLD). Koponen (1995) derived an analytical expression of the characteristic function. This cutoff ensures that the variance will be finite and the distribution converges to a Gaussian in the limit. To model financial prices over time the truncated Lévy flight (TLF) can be constructed by the sum of independent and identically distributed random variables described by a TLD. In contrast

to fat-tailed distributions, the central limit theorem applies and the TLF approaches a Gaussian. Lévy flights have been observed experimentally in physical systems and have been used to describe for instance the spectral random walk of a single molecule embedded in a solid. In these cases an unavoidable cutoff is always present.

In contrast to physical systems, hundreds of papers have shown that there is a strong noni.i.d. clustering effect in financial data. Most of the studies so far performed a summability test of correlated unconditional return data and rejected the Lévy-stable hypothesis. But testing for stable properties requires i.i.d. data. For example Ghose and Kroner (1995) found that summability holds for data generated from an IGARCH process with nonnormal innovations. But of course the simulated data is correlated and not iid. Therefore if we want to justify the use of Lévy processes we first have to remove the autocorrelation in the data and second we can assume the summability of the i.i.d. data. A solution to this problem would be a mechanism to "filter out" the correlations in the data and test if the i.i.d. residual process is adequately described by the stable law.

Fortunately the class of GARCH models has been very successful in modeling the significant volatility clustering and non-i.i.d. nature of the data. More specifically, the standard GARCH model produces a mean reverting time dependent volatility process that "filters out" the correlations in the data and the remaining residuals are assumed to be i.i.d. We develop a specification test based on the augmented GARCH process of Duan (1997) to derive the appropriate model for each data set and each distribution. We show that different choices for the innovations distribution lead to different results about the ability of the GARCH model to explain the autocorrelations in the data. In particular, we show that asymmetry in the volatility is not enough to capture the skewness in the data and a combination of a skewed distribution for the innovations and asymmetry in the volatility process leads to a much better fit in-sample and to better out-of-sample performance.

In general, if the model residuals can be shown to be i.i.d., we can assume the summability of the residuals. If we can assume that the residuals have stable properties we can use the alpha-root-of-time rule for scaling up single day volatility to multi day volatility.

A good parametric description of the distribution of price changes especially in the tails is important for Value at Risk (VaR). Since the Lévy distribution overestimates and the Gaussian underestimates the far tails, we expect the same behavior for the true Value at Risk at high quantiles. To capture the real downside risk, a parametric distribution with fatter tails than the Gaussian and thinner tails than the Lévy should be used. It should be possible to fit the tails separately just to the extreme events. We show that especially for the maximum likelihood method it is crucial to use a skewed distribution to capture the downside risk. For this reason the truncated Lévy distribution is an attractive statistical tool for estimating Value-at-Risk. We use the truncated Lévy flight as a stochastic process for different time intervals and model the time scaling behavior of the quantiles.

The outline of this paper is as follows. In section 2 we will describe the family of Lévystable distributions. Section 3 introduces the idea of a truncation in the tails of the Lévy distribution and summarizes the properties. Section 4 compares the Lévy distribution with a skewed version of the well-known Student t-distribution often used to model fat tails. Section 5 introduces the idea of scale invariant processes and defines the truncated Lévy flight used in this study. Section 6 describes the econometric framework for the study. The empirical results are summarized in section 7. In section 8 we use our models to derive VaR estimates for different time horizons. Finally, section 9 concludes.

2 THE LÉVY DISTRIBUTION

The characteristic function of the four-parameter family of Lévy distributions can be written

$$\psi_{\mathrm{L}}(\mathbf{k}) = i\mu \mathbf{k} - \mathbf{C} |\mathbf{k}|^{\alpha} \{1 + i\beta \operatorname{sgn}(\mathbf{k}) \tan(\pi\alpha/2)\}, \quad 0 < \alpha \le 2.$$

 α is the characteristic exponent determining especially the fatness of the tails, $\beta \in [-1,1]$ is an index of skewness, c is a scale parameter and μ is a location parameter. The density function is only known analytically when $\alpha=1$ (Cauchy distribution) and $\alpha=2$ (Gaussian distribution). However for the symmetric case the value of the density of the Lévy distribution is known at the origin and in the far tails.

In all other situations the density must be generated numerically. Accurate numerical values for the density ψ_L can be calculated Fourier-transforming the CF and evaluating the integral numerically. We use Romberg integration, which allows the prespecification of the tolerated error and in fact a calculation of the density as precise as necessary (see Lambert and Lindsey (1999)). The m-th moment of the distribution can be found from the CF using

$$\langle x^m \rangle = \left(i^m \frac{\partial^m}{\partial k^m} \mathbf{y}_L(k) \right)_{k=0}.$$

The m-th moment is finite only for m< α , for α <2 the variance is infinite. For α =2 all the moments are finite.

3 THE CUTOFF

Modeling financial returns with a distribution with infinite moments is frequently critized. One way to eliminate this problem is to reduce the weight in the tails, because the powerlaw decay with an exponent less than 2 is to slow to ensure that the second moment is finite. A possible solution is to introduce in the tails a function with a faster decay. One possible cutoff is the exponential function for which the CF has been developed (Koponen (1995)). Note the misprint in the original publication, the CF should read:

$$\psi_{\text{TL}}(\mathbf{k}) = C \left\{ \frac{\lambda^{\alpha} - \left(\mathbf{k}^{2} + \lambda^{2}\right)^{\alpha/2}}{\cos(\pi\alpha/2)} \cos\left(\alpha \arctan\left(\frac{|\mathbf{k}|}{\lambda}\right)\right) \left[1 + i \operatorname{sgn}(\mathbf{k})\beta \tan\left(\alpha \arctan\left(\frac{|\mathbf{k}|}{\lambda}\right)\right)\right] \right\}$$

C > 0 is a scale parameter, $0 < \alpha < 2$ (but $\alpha \neq 1$) is the characteristic exponent determining the shape of the distribution and λ is the so-called cutoff parameter, which determines the speed of the decay and as a result the cutoff region. The parameter β determines the skewness when $\beta \neq 0$. The distribution is skewed to the right when $-1 < \beta$ < 0 and skewed to the left when $0 < \beta < 1$. The so-called truncated Lévy distribution shows less weight in the tails and nearly the same shape around the center. Nakao (2000) developed an alternative definition of the exponential truncation of the Lévy distribution. Apart from the scale parameter, the CF is identical to the CF given in Koponen (1995). The truncated Lévy density can also be obtained through an inverse Fourier transform.

For $\mathbf{l} \to +0$ the TLD reduces to the Lévy distribution. In contrast to the Lévy distribution the exponential cutoff ensures that all moments exist. The m-th moment of the distribution can be found from the CF using

$$\left\langle x^{m} \right\rangle = \left(i^{m} \frac{\partial^{m}}{\partial k^{m}} \psi_{TL}(k) \right)_{k=0}$$

The exponential cutoff appears 'earlier' in the tail for liquid markets (higher λ), and can be completely absent in less mature markets.

4 COMPARISON WITH A SKEWED STUDENT-T DISTRIBUTION

A popular distribution in the financial literature is the Student t-distribution. By varying the degrees of freedom it is possible to model every observed tail fatness. The Student t-distribution has the same power law tails as the Lévy distribution in the far tails with the degrees of freedom related to the characteristic exponent. In other words for a value $1<\alpha<2$ equal to the number of degrees of freedom the tails of both distribution are the same but the center is different. In contrast the numbers of degrees of freedom are not restricted. For $\theta>2$ degrees of freedom the distribution behaves as a power-law for large arguments, but at the same time has a finite variance to ensure the validity of the CLT. The density with a scale parameter *a* is defined by:

$$f(u_t, 0, a^2, \boldsymbol{q}) = \frac{1}{\sqrt{\boldsymbol{q}\boldsymbol{p}}} \frac{\Gamma((1+\boldsymbol{q})/2)}{a\Gamma(\boldsymbol{q}/2)} \left[1 + \left(\frac{u_t^2}{a^2\boldsymbol{q}}\right) \right]^{-\left(\frac{\boldsymbol{q}+1}{2}\right)},$$

(-1)

 $(1, \alpha)$

which coincides with the Cauchy distribution for $\theta=1$. In other words for $\alpha=\theta=1$ the Student t-distribution and the Lévy distribution are equal. If $\theta>2$ the variance is defined by:

$$E\left(u_{t}^{2}\right)=\frac{a^{2}\boldsymbol{q}}{\boldsymbol{q}-2}$$

We can rewrite the density function for a t variable with scale parameter σ and derive the Student-t distribution in a standardized version:

$$f(u_t, 0, 1, \boldsymbol{q}) = \frac{1}{\sqrt{\boldsymbol{p}(\boldsymbol{q} - 2)}} \frac{\Gamma((1 + \boldsymbol{q})/2)}{\Gamma(\boldsymbol{q}/2)} \left[1 + \left(\frac{u_t^2}{\boldsymbol{q} - 2}\right) \right]^{-\left(\frac{1 + \boldsymbol{q}}{2}\right)}$$

The t-distribution tends towards a Gaussian in the limit $\theta \rightarrow \infty$.

One feature of the degrees of freedom of the Student t-distribution is that it gives a restriction for the number of moments that exist for the distribution (comparable to the characteristic exponent of the Lévy distribution). A value of θ less than 4 reveals that only the first 3 moments exist. But for financial returns the kurtosis of the sample is well defined. A similar argument was used for a long time against the modeling of financial

returns with the Lévy distribution. As a result there is a restriction that the parameter θ be greater than 4 and at the same time a restriction for the tail fatness.

The Student t distribution has often been used to account for the excess kurtosis in asset returns, but it cannot model the skewness. Paolella (1997) proposed the so-called t_3 distribution, an alternative to the t distribution with skewness. Mittnik et al. (2000) have shown that the skewed distribution outperforms the symmetric distribution in in-sample fit and out-of-sample density predictions.

The standardized density can be written as:

$$f_{t_{3}}(u_{t},0,1,v,d,\boldsymbol{b}) = \frac{1}{K} \begin{cases} \left[1 + \left(\frac{-u_{t}^{d}\boldsymbol{b}^{d}}{v}\right)\right]^{-\left(\frac{1+v}{d}\right)}, & \text{if } u_{t} < 0\\ \left[1 + \left(\frac{u_{t}^{d}\boldsymbol{b}^{-d}}{v}\right)\right]^{-\left(\frac{1+v}{2}\right)}, & \text{if } u_{t} \ge 0 \end{cases},$$

where $K = (\mathbf{b} + \mathbf{b}^{-1})d^{-1}v^{\frac{1}{d}}B(d^{-1}, v)$ and B(.,.) is the beta function. Since the t₃ distribution nests among others the Student-t it can be seen as a generalization of the Student-t distribution. The distribution has power-law tails; therefore *vd* gives an upper limit for the number of moments that exist for the distribution.

5 SCALING PROPERTIES

In practice comparing the distributional properties of price increments at various time intervals provides insight into the temporal dependence structure of the time series. It is possible to reconstruct the distributions for different time intervals from the knowledge of the distribution for short time intervals only if we assume independent and identically distributed (iid) price changes. The Normal and the Lévy distributions plays a central role in this context, because they are stable under addition: the distribution of the sum of a large number of iid random variables belongs to the family of Lévy distributions (generalized Central Limit Theorem) (Gnedenko & Kolmogorov (1954)).

Analyzing the scaling behavior of financial fluctuations is just comparing the increments for shorter time scales τ and for longer time scales N τ . This formally corresponds to summing N random variables. In the case of the Lévy distribution the characteristic function satisfies $N \mathbf{y}_L(k) = \mathbf{y}_L(N^{1/a}k)$ The distribution for various time scales for stationary and independent variables is related by a convolution relation:

$$P_{Nt} = P_t \otimes P_t \otimes \dots \otimes P_t$$

More generally the distribution P(x) of price changes on a time scale $N\tau$ may be obtained from that of a shorter time scale τ by a rescaling of the variable:

$$P_L^N(x) = N^{-1/a} P_L^1(N^{-1/a}x),$$

where $P^{N}(x)$ denotes a N-times convoluted distribution of $P^{1}(x)$.

This relation means that the process x(t) is *self-similar* with a constant self-similar exponent $\frac{1}{a}$. The process is characterized by the scaling behavior of its moments. We can also refer to this as the scaling or fractal property of stochastic processes. The simplest case is the self-similar process with a constant scale factor which is uni-fractal in this sense. (Mandelbrot et al. (1997) introduced the idea of a multifractal process with a time depending random scale factor).

The scenario of a *scale invariant* price process is that of a Lévy flight with a characteristic exponent α which is the inverse of the self-similarity exponent. In other words if the

random variable for short time intervals is distributed according to a Lévy distribution, then in the limit the distribution for long time intervals is also a Lévy with the same characteristic exponent α . The infinite variance of the Lévy distribution prevent that it converges to a Normal distribution.

In Mantegna/Stanley (1995) this scale invariant behavior is observed for short time scales with high frequency data. Cont et al. (1997a) showed that it breaks down for longer time scales. These observations have been explained as a structural break in terms of the truncated Lévy Flight. The scaling behavior is also observed for the variance and the kurtosis of price increments (Cont et al. (1997a)). A link can be established between the scaling behavior of the moments and high-order correlation functions of the time series (Cont (1997b)).

Since we introduced a cut-off for the CF of the truncated Lévy $\mathbf{y}_{TL}(k, \mathbf{l})$, it is no longer self-similar or uni-fractal characterized by the criteria mentioned above, but the process is bi-fractal, the simplest version of a multi-fractal process (see Nakao (2000)). The convolution of the probability distribution can still be obtained by scaling both x and λ . The CF $\mathbf{y}_{TL}(k, \mathbf{l})$ statisfies $N\mathbf{y}_{TL}(k, \mathbf{l}) = \mathbf{y}_{TL}(N^{1/a}k, N^{1/a}\mathbf{l})$ and the N-times convoluted probability distribution satisfies $P_{TL}^N(x, \mathbf{l}) = N^{-1/a}P_{TL}^1(N^{-1/a}x, N^{1/a}\mathbf{l})$.

C is the scale parameter and λ the cutoff-parameter of the so-called truncated Lévy flight (the discrete time alternative to the continuous time process) and therefore both are assumed to grow as $N^{1/a}$. For short time scales (daily) the process behaves like a Lévy flight, but converges towards a Gaussian for longer time scales (say monthly) (see Matacz (2000)). The scaling of λ , which means that for increasing time scales the cutoff is introduced earlier in the tails ensures that the process converges towards a Gaussian process instead of staying a Lévy flight.

To illustrate the RiskMetricsTM approach a Gaussian stochastic process x(t) can be defined by the CF

$$\mathbf{y}_G(k) = -\frac{1}{2}\mathbf{s}^2 t k^2.$$

 σ is the scale parameter of the process and therefore assumed to grow as \sqrt{t} . This is the scaling rule of a usual Gaussian random walk for the standard deviation and used in the standard RiskMetricsTM approach. Our approach can be interpreted as a generalization of the RiskMetricsTM methodology.

6 THE ECONOMETRIC FRAMEWORK

Volatilities can be well predicted with a parametric model such as GARCH. Traditional GARCH models (with Normal- or Student-t distributed error terms) were designed to capture clustering of large and small innovations, which can be modeled as serially correlated conditional variances when the variance exists. In the case of the standard Bollerslev GARCH (p,q) model (Bollerslev (1986)) the conditional variance σ_t^2 can be approximated by

$$\boldsymbol{s}_{t}^{2} = \boldsymbol{a}_{0} + \sum_{i=1}^{p} \boldsymbol{a}_{i} \boldsymbol{s}_{t-i}^{2} + \sum_{j=1}^{q} \boldsymbol{b}_{j} \boldsymbol{s}_{t-j}^{2} (\boldsymbol{e}_{t-j})^{2},$$

where ε_t is a realization from a mean zero, variance one density. The analogue of the standard deviation σ in the family of Lévy distributions is the scale parameter C. If we replace the standard deviation σ by the scale parameter C, we allow C_t to be serially correlated, which produces the volatility clustering.

A more general class of model, which is frequently used together with stable distributions (see Mittnik et al. (1999,2000) is the power-GARCH process of Ding, Granger and Engle (1993):

$$\boldsymbol{s}_{t}^{\boldsymbol{d}} = \boldsymbol{a}_{0} + \sum_{i=1}^{p} \boldsymbol{a}_{i} \boldsymbol{s}_{t-i}^{\boldsymbol{d}} + \sum_{j=1}^{q} \boldsymbol{b}_{j} \boldsymbol{s}_{t-j}^{\boldsymbol{d}} (\left| \boldsymbol{e}_{t-j} \right| - \boldsymbol{r} \boldsymbol{e}_{t-j})^{\boldsymbol{d}}$$

Bollerslev's GARCH model (Bollerslev (1986)) correspond to $\delta=2$ and $\rho=0$. The exponent δ is a parameter of the model and is estimated during the optimization routine. This model specification for p=q=1 leads to a different covariance stationarity condition. In our notation the condition becomes

$$\boldsymbol{a}_1 + E(|\boldsymbol{e}_t| - \boldsymbol{r}\boldsymbol{e}_t)^d \boldsymbol{b}_1 < 1$$

Mittnik et al. (1999) first derived a closed form expression for a symmetric power-GARCH process with Lévy-stable innovations. They show that as δ approaches the characteristic exponent of the Lévy distribution α , $E|\mathbf{e}_t|^d$ increases without bounds and leads to an explosive process, which is not covariance stationary. Most GARCH-stable models in the literature are estimated setting δ equal to α and therefore fail to address the

correct measure of stationarity (the paper by Liu and Brorsen (1995) is the most recent example). The reason for this is that the characteristic exponent α of the Lévy- or the degrees of freedom of the Student-t distribution gives a restriction on the number of moments that exist. Since the stationary condition of an asymmetric power-GARCH process requires the calculation of the fractional moment, $\delta = \alpha$ or $\delta = \theta$ is the limit case and not defined. Apart from the case $\lambda \rightarrow 0$ the moments of the truncated Lévy distribution are finite and the process would be covariance stationary for $\delta = \alpha$. We estimated the model and observed that the alternative definition leads to a different α parameter, because we are combining the power transformation in the volatility process and the parameter of the distribution. We concluded that the two parameters δ and α have to be estimated separately. The motivation for a model with a power transformation of the conditional standard deviation process and the asymmetric absolute residuals lies in the observed autocorrelation functions of $|\mathbf{r}_t|^{\delta}$, see Ding, Granger and Engle (1993) (δ can be interpreted as a separate heteroskedasticity parameter (He and Teräsvirta (1999)).

Another feature of the model is the asymmetric response of volatility to positive and negative shocks, well known as the leverage effect and empirical studies have shown that it is crucial to include leverage parameters (Glosten, Jaganathan and Runkle (1989), Nelson (1991), Ding, Granger and Engle (1993) and Engle and Ng (1992)). Engle and Ng (1993) and Hentschel (1995) studied the asymmetry in terms of the news impact curve. The news impact on volatility can be modeled either with a rotation parameter (like in the power-GARCH or EGARCH) or with a shift parameter (like in the NGARCH). Also a combination of both is possible. In most empirical papers one or the other asymmetric GARCH specification is used without testing if a shift or a rotation is appropriate.

We use a specification test based on the augmented GARCH process of Duan (1997) and determine the type of asymmetry for each data set. Additionally there is no evidence that the appropriate GARCH specification for Gaussian innovations is necessarily the appropriate specification for fat tailed skewed innovation. In particular the asymmetric parameter is assumed to capture the skewness in the data if the innovations distribution is symmetric. Since we are dealing with skewed distributions the relationship between asymmetry in the volatility process and skewness for the innovations distribution is crucial.

For example allowing for skewness in the innovations distribution of the power GARCH process leads to a smaller leverage effect (smaller γ). Therefore we have to test which specification is appropriate for a particular data set and a particular distribution for the innovations.

Duan (1997) developed the augmented GARCH process, a family of parametric GARCH models containing most of the existing GARCH specifications. The augmented GARCH specification can be used to construct a specification test. A Lagrange multiplier test provides an extremely useful class of diagnostic tests. With the augmented GARCH as a general alternative, it can be used to check whether a given GARCH specification is appropriate (see Duan (1997) for details). The LM test deals with local alternatives and suggests, which model is best among the models analyzed. The complete augmented GARCH (1,1) model reads:

$$r_{t} = \hat{i}_{t} + \mathbf{s}_{t} \hat{a}_{t}$$
$$\hat{a}_{t} \sim D(0,1)$$
$$\mathbf{f}_{t} = \mathbf{a}_{0} + \mathbf{g}_{1,t-1} \mathbf{f}_{t-1} + \mathbf{g}_{2,t-1}$$
$$\mathbf{\sigma}_{t} = \begin{cases} |\delta\phi_{t} - \delta + 1|^{1/2\delta} & \text{if } \delta \neq 0 \\ \sqrt{\exp(\phi_{t} - 1)} & \text{if } \delta = 0 \end{cases}$$
$$\mathbf{g}_{1,t-1} = \mathbf{a}_{1} + \mathbf{a}_{2} |\mathbf{e}_{t} - b|^{k} + \mathbf{a}_{3} \max(0, b - \mathbf{e}_{t})^{k}$$
$$\mathbf{g}_{2,t-1} = \mathbf{a}_{4} \frac{|\mathbf{e}_{t} - b|^{k} - 1}{k} + \mathbf{a}_{5} \frac{\max(0, b - \mathbf{e}_{t})^{k} - 1}{k},$$

where the conditional location parameter μ_t can be additionally specified and the conditional scale parameter σ_t is assumed to vary over time. This is a kind of generalization of the original process developed by Duan (1997). Our model is not restricted to zero-mean and variance-one distributions, it also allows for location-zero and scale–one continuous distributions D(0,1). The finite stationary scale parameter and the sufficient condition for strict stationarity of the location adjusted augmented GARCH(1,1) process can be expressed in a general condition. The stationary scale parameter is equal to

$$\overline{\phi} = \frac{\alpha_0 + \alpha_4 E \left[\frac{|\epsilon_t - b|^{\kappa} - 1}{\kappa}\right] + \alpha_5 E \left[\frac{\max(0, b - \epsilon_t)^{\kappa} - 1}{\kappa}\right]}{1 - \alpha_1 - \alpha_2 E[|\epsilon_t - b|^{\kappa}] - \alpha_3 E[\max(0, b - \epsilon_t)^{\kappa}]}$$
$$\overline{\sigma} = \begin{cases} |\delta\overline{\phi} - \delta + 1|^{1/2\delta} & \text{if } \delta \neq 0\\ \sqrt{\exp(\overline{\phi} - 1)} & \text{if } \delta = 0 \end{cases}.$$

Therefore the location-adjusted augmented GARCH(1,1) process is strictly stationary if

$$\alpha_{1} + \alpha_{2} E[|\varepsilon_{t} - b|^{\kappa}] + \alpha_{3} E[\max(0, b - \varepsilon_{t})^{\kappa}] \le 1 \quad \text{if } \delta \neq 0$$
$$\alpha_{1} < 1 \qquad \qquad \text{if } \delta = 0$$

Most GARCH models can be verified as special case of the augmented GARCH process. For some specifications there exist a closed form solution of the unconditional scale parameter or the stationary condition, but otherwise they have to be evaluated numerically. For example for δ =0, κ =1, α_2 =0, α_3 =0, b=0 and α_0 , α_1 , α_4 , α_5 free we derive the EGARCH (1,1) process of Nelson (1991). In contrast to other studies (Hentschel (1995), Franses and van Dijk (1996), Kaiser (1996) and Mittnik et al. (1999)) we experienced no estimation problems with the EGARCH as a special case of the augmented GARCH process.

In the statistical part of the paper we compare three different specifications for the innovations distribution: the Gaussian, a skewed Student-t distribution called t_3 and the skewed truncated Lévy (stable). The GARCH model with normally distributed innovations is nested into both cases with time varying volatility and skewed t_3 *or* truncated Lévy distributed error terms.

All models are estimated with maximum likelihood. The log-likelihood for a series of observations is equal to the sum of the conditional log-likelihood of each observation in the sample. The log-likelihood is given by

$$\ln L = \sum_{t} \left(\ln f(\varepsilon_{t}, 0, 1) - \ln(\sigma_{t}) \right)$$

 $f(\mathbf{e}_{t},0,1)$ is the standardized density and the term $-\ln(\mathbf{s}_{t})$ results from taking the log of the Jacobian of the transformation. The ML estimators follow the standard theory, so they

are consistent and asymptotically normal with mean the parameter estimates and a variance-covariance matrix.

Accurate numerical values for the density of the Lévy distributions can be calculated Fourier-transforming the CF and evaluating the integral numerically. We used Romberg integration, which allows the prespecification of the tolerated error and in fact a calculation of the density as precise as necessary (see Lambert and Lindsey (1999)). Obtaining the density is necessary for the maximum likelihood estimation of the model parameter.

7 EMPIRICAL RESULTS

In this study we use daily closing prices for some major stock market price indices between January 1984 and April 2000. In particular we examine the S&P500, NASDAQ, FTSE 100, HANG SENG and the NIKKEI 225 from May 4, 1992, to April 3, 2000. The total number of trading days covered by the data is between 1955 (NIKKEI 225) and 2001 (S&P500 and NASDAQ). The data are obtained from DataStream.

We used the percentage daily logarithmic change 100 * ln ($p_t/\ p_{t\text{-}1}$), where p_t is the price index at time t.

The summary statistics in table I show that there is skewness and excess kurtosis in the data. The S&P 500, NASDAQ and FTSE 100 show negative skewness related to a fat left tail for the return distribution and the Asian indices show positive skewness related to a fatter right tail. An obvious solution is to use an asymmetric time varying volatility process or a skewed fat tailed distribution. We will show that we need both and that it is not enough to model the excess kurtosis with a fat tailed distribution (Student-t) and trying to capture the skewness with an asymmetric volatility process.

There are some possibilities to model the observed skewness and kurtosis in financial return data. First we can model it by estimating the tails of the empirical distribution separately with a tail estimator or using a skewed fat tailed distribution directly. Second we can model it by fitting an asymmetric time varying volatility process to the data, because the time varying volatility is capturing the excess kurtosis and the asymmetry is capturing the skewness. But the time varying volatility cannot capture all the excess kurtosis and essentially the asymmetry describes the negative correlation between excess returns and volatility; therefore it can reduce the skewness, but it cannot account for all the skewness in the data. Third and more general we can use an *appropriate* asymmetric time-varying volatility process and model the excess kurtosis and skewness separately with a skewed fat tailed distribution. There is already evidence that the inclusion of time-varying volatility (e.g. GARCH) and tail fatness (e.g. Student-t) improve the model. But we have to test if the inclusion of asymmetry and skewed tail fatness improves the model, because both try to model the same phenomenon. In general there are two possibilities to

model the asymmetry: shift or rotation of the news impact curve. Hentschel (1995) studied extensively the news impact curves of various GARCH models and performed a specification test based on the likelihood ratio. In the standard GARCH process the news impact curve has the form of a parabola with a minimum at zero, therefore negative and positive excess returns have the same impact on volatility. If we shift the news impact curve, excess returns with different signs become different impact on volatility. The abovementioned negative correlation between returns and volatility can be modeled with a shift of the news impact curve to the right or a clockwise rotation of the news impact curve. Hentschel showed that the differences in the conditional volatility estimates could be substantially among the various specifications. But how well do skewed distributions work together with asymmetric volatility processes modeled via shift or rotation of the news impact curve? Since the choice of a particular specification will affect the conditional volatility estimate, it will also affect VaR estimates, which is of our particular interest.

In order to determine the appropriate model we make use of a specification test based on the augmented GARCH process developed by Duan (1997), a family of parametric GARCH specification. We conduct a Lagrange multiplier (LM) test, using the augmented GARCH (1,1) as the general alternative. The (robust) LM statistics suggest which of our models analyzed is best.

The parameters estimates of the Gaussian model and a model with skewness and conditional leptokurtosis are reported in table II. We calibrated the models by setting the unconditional scale parameter equal to the sample scale parameter and predetermined α_0 . The results show that our choices of model specifications are motivated by the data.

The significant positive parameters b (NGARCH and power GARCH with shift), α_3 (power GARCH with rotation), α_4, α_5 (EGARCH) and the improved log-likelihood compared to the standard GARCH show generally that there is a leverage effect in index returns. But in particular if we compare the Gaussian and t_3 specifications, the parameters capturing the leverage effect are different and for the t_3 distribution skewness can also be estimated. In general the skewness parameter is reduced if we add an asymmetry parameter to the model. The comparison of the linear GARCH and the asymmetric GARCH models for the t_3 distribution shows that the asymmetric volatility models filter

out some skewness, but there is still skewness in the residuals. For example for the S&P 500 and the NASDAQ we estimated a (significant) skewness parameter and as a result the shift parameter capturing the leverage effect is reduced compared to the Gaussian. The other indices show the same results for a shift in the volatility process, but mixed results for a rotation. In general we can say that if the innovations distribution is significantly skewed to the left, the shift parameter is reduced (NGARCH and power GARCH with shift). For the rotation parameter this is not necessarily true. We can conclude that there's perfect interaction between the skewness parameter of the innovations distribution and the shift parameter in the volatility process, but there is no clear interaction between the skewness parameter.

The LM statistics of all models are presented in table III. It is rather clear from the table that none of the specifications work best in all cases. Nearly each data set needs another GARCH specification to derive the best results. In particular under the hypothesis of conditional normality, a GARCH specification with asymmetry modeled with a rotation of the news impact curve (EGARCH) works best for the Nikkei 225, and the other indices are best described by a shift of the news impact curve (among the models analyzed). For the FTSE 100 we derive no different conclusion under the Hypothesis of a skewed fat tailed distribution. But surprisingly for the other indices the results from the specification test change if we make other assumptions about the innovations distribution. Suddenly a model with rotation is better than the model with shift and vice versa. The S&P 500, NASDAQ and Nikkei 225 are now best described by a power-GARCH with rotation and the description of the Hang Seng index can be improved by introducing a power transformation. These results underline the usefulness of a GARCH specification test, in particular under the hypothesis of skewed fat tailed distributions. Table IV presents the parameter estimates for the appropriate model specifications and the t₃ and truncated Lévy distribution.

Since we model a different scale parameter for each model, we cannot compare them by using likelihood ratio test. But given the appropriate specification we can analyze the residuals of each model by comparing the QQ-plots. Figure II shows the QQ-plots for the residuals of the different models for the whole sample. For illustration we are reporting the results for the NASDAQ index. In general for the power GARCH-N model with shift, which is the appropriate one under conditional normality, the graphs show that the residuals are not normally distributed: there is an underestimation in both tails and a deviation from the empirical distribution in the center. For the model with a fat tailed distribution (Student-t) it can be shown (results not reported), that the fit in the center of the distribution is better, but for every data set the graph shows an underestimation in the left tail and an overestimation in the right tail, because the maximum likelihood method just reports an average degrees of freedom parameter for the whole distribution. Despite the asymmetric volatility parameter in the two models there is still skewness in the residuals. Using a skewed distribution like the t₃ or Lévy distributions should solve this problem. Our approach allows fitting the tails of the distribution separately to the extreme negative and positive events without loosing information in the center of the distribution. The t₃ specification (power-GARCH with rotation) leads to a better fit in both tails compared to the Student-t, but the underestimation in the left tail and the overestimation in the right tail is just reduced. The QQ-plot for the residuals for the truncated Lévy model and the same specification shows that despite the skewness in the data the fit in the tails is extremely precise, there is only a slight overestimation in the tails. The same is not true for the model without cutoff in the tails, because the power law tails of the Lévy distribution lead to an extreme overestimation of the extreme events (results not reported).

From the best performance in-sample we can not necessarily conclude that the model has also superior performance out-of-sample. A backtesting approach will answer the question if the model is also relevant for VaR calculations.

8 IN- AND OUT-OF-SAMPLE VAR ANALYSES

In this section we test the models in- and out-of-sample to see how well they work in practice.

The standard RiskMetricsTM EWMA method assumes that the transformed data, the residuals, are iid standard normally distributed. Both the skewness and excess kurtosis are generally reduce after transformation, but both are still significantly different from zero. It also shows that it is important to incorporate skewness and kurtosis into the model.

First assuming normality we can use the stable property of the Normal distribution and scale the standard deviation at time t by a square-root-of-time rule to get the multi-day values. Second it is well known that we can use this property to estimate the multi-day VaR by multiplying the one-day VaR by \sqrt{T} . Our study shows for a more sophisticated volatility process that it is not possible to guarantee the iid-ness and normality of the residuals. This misspecification results in volatility forecasts using the square-root-of-time scaling rule, which are inappropriate.

Our approach is a generalization of the popular RiskMetricsTM EWMA method, but we are able to provide a better description of the data. First we use the appropriate GARCH model from the specification tests for the particular data set to estimate the time dependent volatility process and to filter out the serial correlations in the data and second we forecast volatility by assuming the truncated Lévy flight for the residuals. We estimate the characteristic exponent α of the truncated Lévy distribution, instead of assuming normality and setting it equal to 2. The process incorporates a α -root-of-time rule as a scaling method for the scale parameter C, instead of assuming normality and using the square-root-of-time rule. Therefore the α -root-of-time rule is the scaling method for the scale parameter λ of the Truncated Lévy Flight and as a result our volatility forecasting method. This allows us to estimate VaR some days ahead.

The very good performance of our model in-sample does not necessary lead to nice outof-sample results. We have to back test our model over a longer period of time.

We want to back test our model for all indices over a period May 1996 until March 2000 and compare it to the RiskMetricsTM EWMA approach. Every day we estimate the model

using the last about 1000 trading days (that means exactly one half of each sample and a moving window) and forecasted the 99% (95%) VaR 5, 10 and 20 days ahead. Table V reports the out-of-sample results. The results are very promising: the VaR estimates we obtained by using our method compared to the RiskMetricsTM method are lower in low volatility periods and higher in high volatility periods, while on average over the whole period our method produces less violations of the expected VaR for all confidence intervals and horizons. The RiskMetricsTM method constantly underpredicts extreme events and this leads very often to an inappropriate number of violations. But in particular this underprediction is slightly reduced for lower confidence intervals or longer forecasting horizons. This is a well-known result for the Gaussian and the square-root-of-time scaling rule. On the other side the GARCH model with the truncated Lévy distribution and the alpha-root-of-time scaling rule leads to an appropriate number of violations for low and high confidence intervals and short and long forecasting horizon. This means not only that the fit in the tails of the distribution is very good, but also that the scaling rule captures the scaling behavior of the data very well and shows a convergence from a skewed leptokurtic distribution to a Gaussian for larger sampling intervals. This is actually the unique bifractal scaling behavior of the truncated Lévy flight.

9 CONCLUSIONS

In this paper we propose a generalization of the popular RiskMetricsTM approach. Our approach is able to capture the observed conditional tail fatness and skewness in financial returns. Using the truncated Lévy flight for the innovations of a GARCH process we propose a new scaling rule to forecast volatility. The in-sample performance of the model as well as the out-of-sample performance for VaR calculations is very satisfactory. There are plenty of practical benefits of this model, which can be explored in future

research.

REFERENCES

- 1. Andersen, T.G. and T. Bollerslev (1997a), "Intraday Periodicity and Volatility Persistence in Financial Markets", *Journal of Empirical Finance*, 4, 115-158.
- Andersen, T. and T. Bollerslev (1998a), "Answering the Skeptics: Yes, Standard Volatility Models Do Provide Accurate Forecasts", *International Economic Review*, 39, 885-905.
- Andersen, T.G., T. Bollerslev, F.X. Diebold and P. Labys (1999), "Forecasting Realized Volatility," work in progress.
- 4. Andersen, T.G., T. Bollerslev and S. Lange (1998), "Forecasting Financial Market Volatility: Sample Frequency vis-à-vis Forecast Horizon," Manuscript, Department of Finance, J.L. Kellogg, Graduate School of Management, Northwestern University, and Departments of Economics, Duke University and University of Virginia.
- Bollerslev, T., (1986), "Generalized Autoregressive Conditional Heteroskedasticity", *Journal of Econometrics*, 31, 307-327.
- Bollerslev, T., R.Y. Chou and K.F. Kroner (1992), "ARCH Modeling in Finance: A Selective Review of the Theory and Empirical Evidence," *Journal of Econometrics*, 52, 5-59.
- Bollerslev, T., R.F. Engle and D.B. Nelson (1994), "ARCH Models," in R.F. Engle and D. McFadden (eds.), *Handbook of Econometrics, Volume IV*, 2959-3038. Amsterdam: North-Holland.
- 8. Bouchaud, J.-P. and M. Potters (2000), "Theory of Financial Risk", *Cambridge University Press*.
- Chong, C.W., Ahmad M. I. and M. Y. Abdullah, (1999), "Performance of GARCH Models in Forecasting Stock Market Volatility", *Journal of Forecasting*, 18, 333-343.
- Christoffersen, P.F. and F.X. Diebold (1998), "How Relevant is Volatility Forecasting for Financial Risk Management?," Wharton Financial Institutions Center Working Paper 97-45 and NBER Working Paper 6844.

- 11. Cont, R., M. Potters and J.-P. Bouchaud (1997a), "Scaling in stock market data: stable laws and beyond", Science & Finance Working Paper (cond-mat/970587).
- 12. Cont, R., (1997b), "Scaling and correlation in financial time series", Science & Finance Working Paper 97-01 (cond-mat/970575).
- Danielsson, J. and C. de Vries (1997), "Value-at-Risk and Extreme Returns", Manuscript, London School of Economics.
- 14. Diebold, F.X., A. Hickman, A. Inoue, T. Schuermann (1997), "Converting 1-Day Volatility to h-Day Volatility: Scaling by h^{1/2} is Worse than You Think", University of Pennsylvania, Working Paper.
- 15. Diebold, F.X., T. Schuermann and A. Inoue (1998), "Scale Models," *RISK*, 11, 104-107.
- Ding, Z., C.W.J. Granger, and R.F. Engle (1993), "A Long Memory Property of Stock Market Returns and a New Model," *Journal of Empirical Finance*, 1, 83-106.
- 17. Drost, F.C. and Nijman, T.E.(1993), "Temporal Aggregation of GARCH Processes", *Econometrica*, 61, 909-927.
- 18. Duan, J. D. (1997), "Augmented GARCH (p,q) process and its diffusion limit", *Journal of Econometrics*, 79, 97-127.
- 19. Engle, R.F. (1982), "Autoregressive Conditional Heteroskedasticity with Estimates of the Variance of UK Inflation," *Econometrica*, 50, 987-1008.
- 20. Engle, R.F. and V. Ng (1993), "Measuring and testing the impact of news on volatility", *Journal of Finance*, 48/5: 1749-78.
- 21. Engle R.F. and J. Mezrich, (1996), "GARCH for Groups", RISK, 36-40.
- 22. Fama, E.F. and R. Roll (1968), "Some Properties of Symmetric Stable Distribution", *Journal of the American Statistical Association*, 63.
- 23. Franses, P.H. and van Dijk, D. (1996). Forecasting Stock Market Volatility Using (Non-Linear) GARCH Models, *Journal of Forecasting*, **15**, 229-235.
- 24. Ghose, D. and K. F. Kroner (1995), "The relationship between GARCH and symmetric stable processes: Finding the source of fat tails in financial data", *Journal of Empirical Finance*, 2, 225-251

- 25. Gnedenko, B.V. & A.N. Kolmogorov, Limit Distributions for Sums of Independent Random Variables, *Addison Wesley*, 1954.
- 26. Granger, C.W.J., Z. Ding and S. Spear (1997), "Stylized Facts on the Temporal and Distributional Properties of Daily Data From Speculative Markets," Manuscript, Department of Economics, University of California, San Diego.
- 27. He, C. and T. Teräsvirta, (1999), "Statistical Properties of the asymmetric power Arch process", In *Cointegration, Causality and Forecasting. Festschrift in honour of Clive W. J. Granger*, Engle R.F., White H. (eds.), Oxford University Press: Oxford.
- Hentschel, L. (1995), "All in the Family: Nesting Symmetric and Asymmetric GARCH Model", *Journal of Financial Economics*, **39**, 71-104.
- 29. Kaiser, T. (1996), "One-Factor-GARCH Models for German Stocks, Estimation and Forecasting", Working Paper, Wirtschaftswissenschaftliche Fakultät, Eberhard Karls Universität Tübingen.
- 30. Koponen, (1995), "Analytic approach to the problem of convergence of truncated Lévy flights towards the Gaussian stochastic process", *Physical Review E*, Volume 52, Number 1, 1197-1199.
- Lambert P. and J.K. Lindsey (1999), "Analysing financial returns by using regression models based on non-symmetric stable distributions", *Journal of Applied Statistics*, 48, 409-424.
- 32. Liu, S. and Brorsen, B.W. (1995), "Maximum Likelihood Estimation of a GARCH-Stable Model", *Journal of Applied Econometrics*, 10, 273-285
- 33. Mantegna, R. N. and H. E. Stanley (1994), Phys. Rev. Lett., 73, 2946
- Mandelbrot, B. (1963), "The Variation of Certain Speculative Prices," *Journal of Business*, 36, 394-419.
- 35. Matacz, A. (2000), "Financial Modeling and Option Theory with the Truncated Lévy Process", *International Journal of Theoretical and Applied Finance*, **3**, 143.
- 36. McCulloch, J.H. (1986), "Simple Consistent Estimator of Stable Distribution Parameters", Communications in Statistics – Simulation and Computation, 15(4), 1109-1136

- 37. McNeil A.J. and R. Frey (1999), "Estimation of Tail-Related Risk Measures for Heteroskedastic Financial Time Series: an Extreme Value Approach", Manuscript, University of Zurich
- 38. Mittnik, S. Paolella, M.S., Rachev, S.T. (1999), "The Prediction of Down-side Market Risk with GARCH-stable Models", Working Paper 118, Institute of Statistics and Econometrics, University of Kiel
- 39. Mittnik, S. and M. S. Paolella, (2000), "Conditional Density and Value-at-Risk Prediction of Asian Currency Exchange Rates", *Journal of Forecasting*, **19**, 313-333
- 40. Nakao, H., (2000), "Multi-scaling properties of truncated Lévy flights", Manuscript, Graduate School of Mathematical Sciences, University of Tokyo.
- Pownall, R. A.J. and Koedijk, K.G. (1999), "Capturing downside risk in financial markets: the case of the Asian Crisis, *Journal of International Money and Finance*, 18, 853-870.
- 42. RiskMetricsTM (1996), J.P. Morgan, "Technical document", fourth edition.
- 43. Schwert, G.W. (1989), "Why does Stock Market Volatility Change over Time?," *Journal of Finance*, 44, 1115-1154.
- 44. Taylor, S. (1986), "Modeling Financial Time Series", John Wiley, New York.
- 45. Yang S.R. and B.W. Brorsen (1995), "Nonlinear dynamics of daily foreign exchange rates", *Advances in Quantitative Finance and Accounting*.

able I: In-Sample Analysis: Parameter estimates										
	Parameter	S&P 500	NASDAQ	FTSE 100	HANG SENG	NIKKEI 225				
SUMMARY	r didiriotor									
STATISTICS	MEAN	0.064	0.098	0.044	0.056	0.007				
	ST, DEV,	0.907	1.228	0.942	1.889	1.430				
	MINIMUM	-7.113	-8.954	-4.140	-14.735	-5.957				
	MAXIMUM	4.989	5.848	4.345	17.247	7.660				
	SKEWNESS	-0.398	-0.676	-0.055	0.097	0.200				
	KURTOSIS	9.224	7.942	4.748	11.870	5.462				
	# OF OBSERVATIONS	2001	2001	2000	1964	1955				

Index	Parameter	GARCH	Power GARCH (rotation)	NGARCH	EGARCH	Power GARCH (shift)
6 ° D 500						
N (0,1)	γ (SE)	1.000	0.755	1.000	0.000	0.902
	(SE)	(0.000)	(0.000)	(0.000)	(0.000)	(0.108)
	к (ог)	2.000	1.511	2.000	1.000	1.804
	(SE) (SE)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	α1	0.940	0.907	0.898	0.984	0.888
	(SE) (SE)	(0.006) (0.020)	(0.011) (0.032)	(0.008) (0.042)	(0.002) (0.008)	(0.010) (0.044)
	α2	0.054	0.026	0.064	0.000	0.083
	(SE) (SE)	(0.005)	(0.016)	(0.007)	(0.000)	(0.013) (0.037)
	α3	0.000	0.149	0.000	0.000	0.000
	(SE)	(0.000)	(0.021)	(0.000)	(0.000)	(0.000)
	(SE) α4	0.000	0.000	0.000	0.055	0.000
	(SE)	(0.000)	(0.000)	(0.000)	(0.019)	(0.000)
	(SE) α5	(0.000)	0.000)	0.000)	(0.018)	(0.000)
	(SE)	(0.000)	(0.000)	(0.000)	(0.018)	(0.000)
	(SE) b	(0.000)	(0.000)	(0.000)	(0.055)	(0.000)
	(SE)	(0.000)	(0.000)	(0.126)	(0.000)	(0.115)
	(SE)	(0.000)	(0.000)	(0.118)	(0.000)	(0.114)
	Log-Likelihood	-2376.8	-2354.5	-2359.5	-2354.6	-2358.7
t3 (0,1,v,d, b)	γ	1.000	0.663	1.000	0.000	0.842
	(SE)	(0.000)	(0.100)	(0.000)	(0.000)	(0.105)
	(SE) K	(0.000)	(0.153)	(0.000)	(0.000)	(0.153) 1.685
	(SE)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	(SE)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	(SE)	(0.008)	(0.019)	(0.010)	(0.003)	(0.017)
	(SE)	(0.015)	(0.025)	(0.025)	(0.006)	(0.030)
	(SE)	(0.010)	(0.030)	(0.011)	(0.000)	(0.092
	(SE)	(0.016)	(0.028)	(0.015)	(0.000)	(0.038)
	α.3 (SE)	0.000 (0.000)	0.157 (0.039)	0.000 (0.000)	0.000 (0.000)	0.000 (0.000)
	(SE)	(0.000)	(0.049)	(0.000)	(0.000)	(0.000)
	α4 (SE)	0.000	0.000	0.000	0.057	0.000
	(SE)	(0.000)	(0.000)	(0.000)	(0.018)	(0.000)
	α5 (SE)	0.000	0.000	0.000	0.154	0.000
	(SE)	(0.000)	(0.000)	(0.000)	(0.052)	(0.000)
	b (OF)	0.000	0.000	0.532	0.000	0.457
	(SE) (SE)	(0.000)	(0.000)	(0.172) (0.134)	(0.000)	(0.148) (0.145)
	V	5.566	5.343	6.031	5.263	5.866
	(SE) (SE)	(0.171) (0.178)	(2.447) (2.534)	(2.637) (3.168)	(2.435) (2.595)	(2.677) (3.093)
1	d	1.715	1.733	1.702	1.727	1.709
1	(SE) (SE)	(0.026)	(0.179)	(0.166) (0.184)	(0.179) (0.177)	(0.172) (0.183)
	β	0.975	1.008	1.003	0.999	1.000
	(SE)	(0.025)	(0.027)	(0.027)	(0.027)	(0.027)
	(3E)	(0.026)	(0.018)	(0.023)	(0.022)	(0.025)

		GARCH	Power GARCH	NGARCH	EGARCH	Power GARCH
Index	Parameter	OARON	(rotation)	NOARON	LOARON	(shift)
			((0)
Nasdaq						
N (0,1)	γ	1.000	0.675	1.000	0.000	0.770
	(SE)	(0.000)	(0.083)	(0.000)	(0.000)	(0.088)
	(SE)	2,000	(0.080)	(0.000)	(0.000)	(0.085)
	K (SE)	2.000 (0.000)	1.350 (0.000)	2.000 (0.000)	1.000 (0.000)	1.540
	(SE)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	α1	0.859	0.818	0.828	0.976	0.805
	(SE)	(0.012)	(0.025)	(0.015)	(0.004)	(0.023)
	(SE)	(0.045)	(0.050)	(0.056)	(0.014)	(0.049)
	α2 (SE)	0.123	0.130	0.130	0.000	0.109
	(SE)	(0.036)	(0.032)	(0.036)	(0.000)	(0.033)
	α3	0.000	0.124	0.000	0.000	0.000
	(SE)	(0.000)	(0.015)	(0.000)	(0.000)	(0.000)
	(SE)	(0.000)	(0.061)	(0.000)	(0.000)	(0.000)
	α4 (SE)	0.000	0.000	0.000	0.182	0.000
	(SE)	(0.000)	(0.000)	(0.000)	(0.024)	(0.000)
	α5	0.000	0.000	0.000	0.117	0.000
	(SE)	(0.000)	(0.000)	(0.000)	(0.016)	(0.000)
	(SE)	(0.000)	(0.000)	(0.000)	(0.053)	(0.000)
	b (CE)	0.000	0.000	0.311	0.000	0.291
	(SE) (SE)	(0.000)	(0.000)	(0.047)	(0.000)	(0.044)
	(02)	(0.000)	(0.000)	(0.007)	(0.000)	(0.001)
	Log-Likelihood	-2888.7	-2872.2	-2875.5	-2876.2	-2872.0
t3 (0,1,v,d, b)	γ	1.000	0.590	1.000	0.000	0.614
	(SE)	(0.000)	(0.119)	(0.000)	(0.000)	(0.126)
	(SE)	2 000	(0.086)	2 000	(0.000)	(0.093)
	(SE)	2.000 (0.000)	(0.000)	2.000 (0.000)	(0.000)	(0.000)
	(SE)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	α1	0.908	0.864	0.905	0.993	0.859
	(SE)	(0.012)	(0.028)	(0.012)	(0.004)	(0.028)
	(SE)	(0.036)	(0.040)	(0.039)	(800.0)	(0.042)
	(SE)	(0.015)	(0.048)	(0.017)	0.000 (0.000)	(0.198
	(SE)	(0.033)	(0.043)	(0.033)	(0.000)	(0.050)
	α3	0.000	0.060	0.000	0.000	0.000
	(SE)	(0.000)	(0.032)	(0.000)	(0.000)	(0.000)
	(SE)	(0.000)	(0.037)	(0.000)	(0.000)	(0.000)
	α4 (SE)	0.000 (0.000)	0.000 (0.000)	0.000 (0.000)	0.201	0.000
	(SE)	(0.000)	(0.000)	(0.000)	(0.050)	(0.000)
	α5	0.000	0.000	0.000	0.060	0.000
	(SE)	(0.000)	(0.000)	(0.000)	(0.035)	(0.000)
	(SE)	(0.000)	(0.000)	(0.000)	(0.039)	(0.000)
	D (SE)	0.000 (0.000)	0.000	0.101 (0.084)	0.000	0.088
	(SE)	(0.000)	(0.000)	(0.088)	(0.000)	(0.058)
	v	6.592	5.773	6.260	5.059	6.190
	(SE)	(0.189)	(2.859)	(3.106)	(2.377)	(3.208)
	(SE)	(0.248)	(3.429)	(4.054)	(2.646)	(3.785)
	(SE)	1.000	1.860 (0.104)	1.00/	1.933	1.835
	(SE)	(0.021)	(0.237)	(0.191)	(0.227)	(0.229)
	β	0.877	0.889	0.890	0.879	0.883
	(SE)	(0.030)	(0.023)	(0.023)	(0.023)	(0.023)
	(SE)	(0.034)	(0.025)	(0.023)	(0.021)	(0.022)
	l og l ikelihood	-2838 5	-2830.8	-2837 6	-2831.6	-2831 7
		2000.0	-2000.0	2001.0	2001.0	2001.7

Index	Parameter	GARCH	Power GARCH (rotation)	NGARCH	EGARCH	Power GARCH (shift)
Index	T diameter		(iotation)			(Shirt)
Ftse 100						
N (0,1)	γ	1.000	0.673	1.000	0.000	0.756
	(SE)	(0.000)) (0.131)	(0.000)	(0.000)	(0.141)
	(SE)	0.000) (0.168)	(0.000)	(0.000)	(0.147)
	K (SE)	2.000	1.346	2.000	1.000	1.512
	(SE)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	α1	0.951	0.930	0.944	0.993	0.921
	(SE)	(0.007) (0.019)	(0.006)	(0.002)	(0.017)
	(SE)	(0.011) (0.022)	(0.014)	(0.003)	(0.028)
	α2	0.044	0.044	0.039	0.000	0.070
	(SE)	(0.006	(0.026)	(0.006)	(0.000)	(0.026)
	(SE)	0.009) (0.022)	(800.0)	(0.000)	(0.029)
	α.3 (SE)	0.000	0.069	0.000	0.000	0.000
	(SE)	(0.000)	(0.020)	(0.000)	(0.000)	(0.000)
	α4	0.000	0.000	0.000	0.056	0.000
	(SE)	(0.000) (0.000)	(0.000)	(0.020)	(0.000)
	(SE)	(0.000	(0.000)	(0.000)	(0.016)	(0.000)
	α.5	0.000	0.000	0.000	0.070	0.000
	(SE)	(0.000) (0.000)	(0.000)	(0.018)	(0.000)
	(SE)	(0.000)) (0.000)	(0.000)	(0.024)	(0.000)
	D (OF)	0.000	0.000	0.561	0.000	0.544
	(SE)	(0.000)) (0.000)	(0.159)	(0.000)	(0.152)
	(02)	(0.000	(0.000)	(0.100)	(0.000)	(0.101)
	Log-Likelihood	-2525.2	-2517.3	-2516.6	-2520.0	-2515.2
(0.1)(d.h)	~	1 000	0 707	1 000	0.000	0.600
i3 (0, 1, v, u, b)	(SF)	0.000	(0.161)	(0.000)	(0.000)	0.090 (0.166)
	(SE)	(0.000)	(0.176)	(0.000)	(0.000)	(0.200)
	κ	2.000	1.414	2.000	1.000	1.380
	(SE)	(0.000) (0.000)	(0.000)	(0.000)	(0.000)
	(SE)	(0.000) (0.000)	(0.000)	(0.000)	(0.000)
	α1	0.950	0.944	0.941	0.995	0.923
	(SE)	(0.009) (0.014)	(0.009)	(0.003)	(0.017)
	(SE) α2	0.082	0.043	0.071	0.003)	(0.022)
	(SF)	(0.002	(0.043	(0.013)	(0.000)	(0.028)
	(SE)	(0.015) (0.021)	(0.012)	(0.000)	(0.031)
	α3	0.000	0.096	0.000	0.000	0.000
	(SE)	(0.000) (0.027)	(0.000)	(0.000)	(0.000)
	(SE)	(0.000) (0.030)	(0.000)	(0.000)	(0.000)
	α4	0.000	0.000	0.000	0.069	0.000
	(SE)	(0.000) (0.000)	(0.000)	(0.029)	(0.000)
	(SE)	0.000	0.000	0.000	0.108	0.000)
	(SF)	0.000 (0.000	0.000	0.000	(0.030)	0.000 (0.000)
	(SE)	(0.000) (0.000)	(0.000)	(0.031)	(0.000)
	b	0.000	0.000	0.438	0.000	0.407
	(SE)	(0.000) (0.000)	(0.135)	(0.000)	(0.116)
	(SE)	(0.000) (0.000)	(0.120)	(0.000)	(0.074)
	V (OE)	4.469	3.638	3.648	3.741	3.551
	(SE) (SE)	(0.285)	(1.582)	(1.488) (1.290)	(1.699)	(1.454)
	d (3L)	2.346	2.437	2.471	2.416	2.472
	(SE)	(0.026) (0.292)	(0.287)	(0.292)	(0.291)
	(SE)	(0.023	(0.285)	(0.283)	(0.266)	(0.288)
	β	0.938	0.976	0.980	0.969	0.972
	(SE)	(0.035	(0.028)	(0.028)	(0.027)	(0.028)
	(SE)	(0.034) (0.030)	(0.028)	(0.027)	(0.030)
	l og l ikeliheed	251E 0	2509.0	2507 4	2510.0	250F 7
	LUG-LIKEIIII000	-2010.0	-2000.0	-2007.1	-2010.0	-2000.7

	Table II: In-Sam	ple Analysis	: Estimates	of the scale	parameter	equation
--	------------------	--------------	-------------	--------------	-----------	----------

Index	Parameter	GARCH	Power GARCH (rotation)	NGARCH	EGARCH	Power GARCH (shift)
Hang Sang						
N (0 1)	24	1 000	0.686	1 000	0.000	0.882
N (0, 1)	y (SE)	(0.000)	0.000	(0.000)	(0.000	(0.123)
	(SE)	(0.000)	(0.087)	(0.000)	(0.000)	(0.123)
	(02)	2 000	1 372	2 000	1 000	1 764
	(SE)	2.000 (0.000)	(0.000)	(0.000)	(0.000)	(0,000)
	(SE)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	α1	0.887	0.883	0.876	0.977	0.875
	(SE)	(0.011)	(0.012)	(0.010)	(0.003)	(0.011)
	(SE)	(0.021)	(0.018)	(0.021)	(0.007)	(0.021)
	α2	0.099	0.070	0.094	0.000	0.106
	(SE)	(0.009)	(0.015)	(0.009)	(0.000)	(0.016)
	(SE)	(0.017)	(0.018)	(0.013)	(0.000)	(0.019)
	α3	0.000	0.100	0.000	0.000	0.000
	(SE)	(0.000)	(0.013)	(0.000)	(0.000)	(0.000)
	(SE)	(0.000)	(0.028)	(0.000)	(0.000)	(0.000)
	α4	0.000	0.000	0.000	0.129	0.000
	(SE)	(0.000)	(0.000)	(0.000)	(0.020)	(0.000)
	(SE)	(0.000)	(0.000)	(0.000)	(0.023)	(0.000)
	α5	0.000	0.000	0.000	0.128	0.000
	(SE)	(0.000)	(0.000)	(0.000)	(0.018)	(0.000)
	(SE)	(0.000)	(0.000)	(0.000)	(0.034)	(0.000)
	b	0.000	0.000	0.393	0.000	0.371
	(SE)	(0.000)	(0.000)	(0.065)	(0.000)	(0.064)
	(SE)	(0.000)	(0.000)	(0.110)	(0.000)	(0.123)
	Log-Likelihood	-3709.3	-3688.6	-3696.7	-3689.0	-3696.2
t3 (0,1,v,d, b)	γ	1.000	0.630	1.000	0.000	0.840
	(SE)	(0.000)	(0.154)	(0.000)	(0.000)	(0.185)
	(SE)	(0.000)	(0.097)	(0.000)	(0.000)	(0.121)
	ĸ	2.000	1.260	2.000	1.000	1.681
	(SE)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	(SE)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	(CE)	0.901	0.092	0.000	0.900	0.002
	(SE)	(0.013)	(0.018)	(0.016)	(0.004)	(0.017)
	(OL)	0.104	0.083	0.106	0.000	0.123
	(SE)	(0.010)	(0.028)	(0.019)	(0.000	(0.030)
	(SE)	(0.013)	(0.020)	(0.013)	(0.000)	(0.030)
	α3 (°-)	0.000	0 114	0.000	0.000	0.000
	(SE)	(0.000)	(0.027)	(0.000)	(0.000)	(0.000)
	(SE)	(0.000)	(0.034)	(0.000)	(0.000)	(0.000)
	α4	0.000	0.000	0.000	0.140	0.000
	(SE)	(0.000)	(0.000)	(0.000)	(0.034)	(0.000)
	(SE)	(0.000)	(0.000)	(0.000)	(0.025)	(0.000)
	α5	0.000	0.000	0.000	0.132	0.000
	(SE)	(0.000)	(0.000)	(0.000)	(0.034)	(0.000)
	(SE)	(0.000)	(0.000)	(0.000)	(0.039)	(0.000)
	b	0.000	0.000	0.408	0.000	0.375
	(SE)	(0.000)	(0.000)	(0.111)	(0.000)	(0.105)
	(SE)	(0.000)	(0.000)	(0.105)	(0.000)	(0.122)
	V	17.510	18.498	14.369	12.060	14.838
	(SE)	(10.145)	(22.798)	(14.768)	(11.187)	(15.367)
	(SE)	(11.158)	(26.439)	(17.864)	(12.371)	(18.262)
	d (OF)	1.428	1.447	1.457	1.543	1.452
	(SE)	(0.123)	(0.148)	(0.153)	(0.162)	(0.151)
	(SE)	(0.124)	(0.156)	(0.167)	(0.100)	0.002
	p (SE)	0.970	0.330	0.330	0.991	0.993
	(SE)	(0.046)	(0.025)	(0.025)	(0.025)	(0.026)
	(01)	(0.000)	(0.020)	(0.022)	(0.017)	(0.021)
	l og-l ikelihood	-3659 5	-3646 1	-3651.0	-3651 4	-3650.5
	Log Lincillioou	0000.0	00-10.1	0001.0	0001.4	0000.0

Table II: In-Sample Analys	is: Estimates	of the s	cale parameter	equation								
Index	Parame	ter	GARCH		Power GARCH (rotation)	1	NGARCH		EGARCH		Power GAR((shift)	СН
Nikkoj 225												
N (0,1)	γ		1.000		0.613		1.000		0.000		0.390	
		(SE)		(0.000)	((0.117)		(0.000)		(0.000)		(0.076)
	r	(SE)	2 000	(0.000)	1 227	(0.225)	2 000	(0.000)	1 000	(0.000)	0 780	(0.161)
	i.	(SE)	2.000	(0.000)	((0.000)	2.000	(0.000)	1.000	(0.000)	0.100	(0.000)
		(SE)	0.070	(0.000)	((0.000)	0.000	(0.000)	0.000	(0.000)	0.774	(0.000)
	αΙ	(SE)	0.870	(0.014)	0.867	(0.015)	0.863	(0.015)	0.962	(0.004)	0.774	(0.026)
		(SE)		(0.035)	((0.058)		(0.036)		(0.023)		(0.108)
	α2	(05)	0.090	(0.040)	0.079	(0.045)	0.086	(0,000)	0.000	(0.000)	0.214	(0.004)
		(SE) (SE)		(0.010) (0.019)	((0.015) (0.070)		(0.009) (0.025)		(0.000)		(0.031) (0.106)
	α3	(02)	0.000	(0.0.0)	0.087	()	0.000	(0.020)	0.000	()	0.000	()
		(SE)		(0.000)	((0.016)		(0.000)		(0.000)		(0.000)
	α4	(SE)	0.000	(0.000)	0.000	(0.045)	0.000	(0.000)	0 116	(0.000)	0.000	(0.000)
	4	(SE)	0.000	(0.000)	0.000 ((0.000)	0.000	(0.000)	0.110	(0.012)	0.000	(0.000)
	_	(SE)		(0.000)	((0.000)		(0.000)		(0.070)		(0.000)
	α5	(SE)	0.000	(0,000)	0.000	(0.000)	0.000	(0,000)	0.122	(0.017)	0.000	(0.000)
		(SE)		(0.000)	((0.000)		(0.000)		(0.056)		(0.000)
	b	(0-)	0.000	(0.000)	0.000	(0.000)	0.426	(0.0.40)	0.000	(0.000)	0.527	(0.000)
		(SE) (SE)		(0.000)	((0.000)		(0.046) (0.301)		(0.000)		(0.033) (0.014)
	المع النادمانا	,	2206.0	()	2295 7	()	2207.0	()	2202.2	()	2280.4	()
	Log-Likeli	1000	-3396.9		-3385.7		-3387.8		-3383.2		-3380.1	
t3 (0,1,v,d, b)	γ	(05)	1.000	(0,000)	0.725	(0.450)	1.000	(0,000)	0.000	(0.000)	0.723	(0.400)
		(SE) (SE)		(0.000)	((0.153)		(0.000)		(0.000)		(0.168)
	κ	()	2.000	()	1.451	(•••••)	2.000	()	1.000	()	1.446	()
		(SE)		(0.000)	((0.000)		(0.000)		(0.000)		(0.000)
	α1	(SE)	0.919	(0.000)	0.919	(0.000)	0.897	(0.000)	0.986	(0.000)	0.885	(0.000)
		(SE)		(0.013)	((0.016)		(0.015)		(0.005)		(0.019)
	~?	(SE)	0.000	(0.014)	0.040	(0.014)	0.070	(0.017)	0.000	(0.005)	0 1 1 7	(0.024)
	u2	(SE)	0.090	(0.015)	0.040	(0.021)	0.079	(0.013)	0.000	(0.000)	0.117	(0.030)
		(SE)		(0.015)	Ì	(0.016)		(0.013)		(0.000)		(0.036)
	α.3	(SE)	0.000	(0,000)	0.138	(0.029)	0.000	(0,000)	0.000	(0,000)	0.000	(0,000)
		(SE)		(0.000)	((0.028)		(0.000)		(0.000)		(0.000)
	α4		0.000		0.000		0.000		0.077		0.000	
		(SE)		(0.000)	((0.000)		(0.000)		(0.028)		(0.000)
	α5	(0L)	0.000	(0.000)	0.000	(0.000)	0.000	(0.000)	0.184	(0.020)	0.000	(0.000)
		(SE)		(0.000)	((0.000)		(0.000)		(0.034)		(0.000)
	h	(SE)	0.000	(0.000)	0.000	(0.000)	0.683	(0.000)	0.000	(0.036)	0.600	(0.000)
	5	(SE)	0.000	(0.000)	0.000 ((0.000)	0.000	(0.136)	0.000	(0.000)	0.000	(0.112)
		(SE)	0.747	(0.000)	((0.000)	0.544	(0.152)	0.400	(0.000)	0.447	(0.012)
	v	(SE)	2.747	(0.198)	2.350	(0.521)	2.541	(0.538)	2.196	(0.506)	2.417	(0.532)
		(SE)		(0.265)	Ì	(0.794)		(0.832)		(0.701)		(0.839)
	d	(SE)	2.120	(0.025)	2.256	(0.217)	2.230	(0.205)	2.300	(0.220)	2.255	(0.215)
		(SE) (SE)		(0.025) (0.024)	((0.217)		(0.205) (0.283)		(0.230) (0.281)		(0.215) (0.298)
	β	. ,	0.956		1.014		1.016		1.004	. ,	1.007	. ,
		(SE)		(0.047)	((0.027)		(0.027)		(0.028)		(0.028)
		(31)		(0.040)	(0.029)		(0.029)		(0.027)		(0.020)
	Log-Likeli	nood	-3300.6		-3284.5		-3285.4		-3284.1		-3284.6	

Index		GARCH	Power GARCH (rotation)	NGARCI	H	EGARCH		Power GAF (shift)	RCH
S&P 500									
N (0,1)	LM (p-value) Robust LM (p-value)	42.9 (0.0000) 41.9 (0.0000)	17.8 (0.00 ⁻ 9.9 (0.04	10.9 3) 9.0	(0.0542) (0.1075)	22.1 20.3	(0.0005) (0.0011)	9.8 5.5	(0.0439) (0.2362)
t3 (0,1,v,d, b) truncated stable (0,1, a , 1 , b)	LM (p-value) Robust LM (p-value)	37.4 (0.0000) 33.3 (0.0000)	9.7 (0.04 4.9 (0.30	19.0 (4) 11.0	(0.0019) (0.0504)	13.4 10.5	(0.0198) (0.0613)	21.4 8.3	(0.0003) (0.0799)
Nasdaq N (0,1)	LM (p-value) Robust LM (p-value)	35.1 (0.0000) 33.2 (0.0000)	1.2 (0.88) 1.0 (0.90)	10.8 5) 9.9	(0.0544) (0.0779)	15.8 15.4	(0.0076) (0.0089)	0.8 0.7	(0.9401) (0.9472)
t3 (0,1,v,d, b) truncated stable (0,1, a , 1 , b)	LM (p-value) Robust LM (p-value)	19.6 (0.0032) 16.6 (0.0110)	2.1 (0.724 1.6 (0.805	17.8 14.7 14.7	(0.0032) (0.0119)	8.3 8.1	(0.1415) (0.1482)	3.9 2.7	(0.4206) (0.6159)
Ftse 100 N (0,1)	LM (p-value) Robust LM (p-value)	20.8 (0.0020) 18.4 (0.0054)	5.7 (0.22 4.6 (0.32)	3.6 4) 3.0	(0.6064) (0.7041)	17.7 17.4	(0.0034) (0.0038)	4.6 4.0	(0.3354) (0.4090)
t3 (0,1,v,d, b) truncated stable (0,1, a ,1, b)	LM (p-value) Robust LM (p-value)	26.3 (0.0002) 19.8 (0.0031)	17.7 (0.00 11.7 (0.01	8.5 4) 6.2	(0.1291) (0.2858)	12.8 11.7	(0.0249) (0.0387)	6.6 5.1	(0.1591) (0.2727)
Hang Seng N (0, 1) t3 (0, 1, v, d, b)	LM (p-value) Robust LM (p-value) LM	43.6 (0.0000) 38.0 (0.0000) 43.3	13.0 (0.01) 12.3 (0.01) 19.0	5) 14.6 10.1 19.4	(0.0120) (0.0736)	25.2 24.2 23.5	(0.0001) (0.0002)	14.4 8.8 18.7	(0.0060) (0.0662)
truncated stable (0,1, a,1,b)	(p-value) Robust LM (p-value)	(0.0000) 33.2 (0.0000)	(0.00) 15.7 (0.00)	10.9 (5)	(0.0016) (0.0534)	21.7	(0.0003) (0.0006)	8.3	(0.0009)
Nikkei 225 N (0,1)	LM (p-value) Robust LM (p-value)	22.2 (0.0011) 16.8 (0.0102)	7.7 (0.10 4.9 (0.29	11.1 8) 8.4 64)	(0.0497) (0.1360)	6.4 4.3	(0.2709) (0.5008)	6.5 5.2	(0.1652) (0.2689)
t3 (0,1,v,d, b) truncated stable (0,1, a,1,b)	LM (p-value) Robust LM (p-value)	35.5 (0.0000) 31.1 (0.0000)	1.9 (0.75- 1.3 (0.86)	9.1 0) 7.4	(0.1050) (0.1924)	6.5 6.3	(0.2641) (0.2769)	3.0 1.8	(0.5500) (0.7760)

		Developmenter										
Index	Log-Likelihood	Parameter γ	к	α1	α2	α3	α4	α5	b	v / α	d/λ	β
S&P 500												
POWER GARCH, ROTATION												
t ₃ (0, 1, v, d, b)	-2303.5	0.6626 (0.1002) (0.1532)	1.3253 (0.0000) (0.0000)	0.9168 (0.0193) (0.0254)	0.0383 (0.0297) (0.0275)	0.1570 (0.0388) (0.0493)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	5.3433 (2.4472) (2.5337)	1.7325 (0.1788) (0.1695)	1.0078 (0.0266) (0.0177)
truncated stable (0,1, a , l , b)	-2304.5	0.8802 (0.0334) (0.0577)	1.7604 (0.0000) (0.0000)	0.9270 (0.0146) (0.0180)	0.0080 (0.0091) (0.0069)	0.0725 (0.0197) (0.0258)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	1.6985 (0.1049) (0.1846)	0.2349 (0.1166) (0.1531)	0.2756 (0.1327) (0.2483)
Nasdaq												
POWER GARCH, ROTATION												
t ₃ (0, 1, v, d, b)	-2830.8	0.5898 (0.1189) (0.0864)	1.1796 (0.0000) (0.0000)	0.8643 (0.0276) (0.0402)	0.1607 (0.0480) (0.0428)	0.0597 (0.0322) (0.0371)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	5.7726 (2.8585) (3.4286)	1.8600 (0.1936) (0.0000)	0.8885 (0.0230) (0.0251)
truncated stable (0,1, a , l , b)	-2815.3	0.790842 (0.0519) (0.0488)	1.5817 (0.0000) (0.0000)	0.876852 (0.0220) (0.0288)	0.040436 (0.0146) (0.0116)	0.070832 (0.0216) (0.0220)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	1.813647 (0.0445) (0.0362)	0.088733 (0.0410) (0.0459)	0.918438 (0.1407) (0.1200)
Ftse 100												
NGARCH, SHIFT												
t ₃ (0,1,v,d, b)	-2507.1	1.0000 (0.0000) (0.0000)	2.0000 (0.0000) (0.0000)	0.9411 (0.0088) (0.0110)	0.0713 (0.0134) (0.0120)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	0.4381 (0.1353) (0.1198)	3.6484 (1.4876) (1.3795)	2.4710 (0.2874) (0.2828)	0.9800 (0.0282) (0.0282)
truncated stable (0,1, a , l , b)	-2505.3	1.0000 (0.0000) (0.0000)	2.0000 (0.0000) (0.0000)	0.9348 (0.0101) (0.0113)	0.0188 (0.0035) (0.0031)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	1.1163 (0.2745) (0.2420)	1.9057 (0.0653) (0.0548)	0.1639 (0.1570) (0.1171)	0.4193 (0.2889) (0.2889)
Hang Seng												
POWER GARCH, SHIFT												
t ₃ (0, 1, v, d, b)	-3650.5	0.8403 (0.1847) (0.1209)	1.6806 (0.0000) (0.0000)	0.8817 (0.0171) (0.0227)	0.1226 (0.0301) (0.0219)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	0.3750 (0.1050) (0.1219)	14.8378 (15.3670) (18.2621)	1.4524 (0.1506) (0.1618)	0.9934 (0.0256) (0.0210)
truncated stable (0,1, a,1,b)	-3659.3	0.7260 (0.1212) (0.0750)	1.4520 (0.0000) (0.0000)	0.8876 (0.0195) (0.0303)	0.0642 (0.0233) (0.0174)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	0.6443 (0.1595) (0.2598)	1.6208 (0.1133) (0.1116)	0.2401 (0.1411) (0.0961)	0.0064 (0.0773) (0.0804)
Nikkei 225												
POWER GARCH, ROTATION												
t ₃ (0,1,v,d, b)	-3284.5	0.7254 (0.1533) (0.1141)	1.4508 (0.0000) (0.0000)	0.9186 (0.0156) (0.0136)	0.0403 (0.0214) (0.0164)	0.1380 (0.0276) (0.0293)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	2.3501 (0.5206) (0.7943)	2.2564 (0.2167) (0.2920)	1.0142 (0.0269) (0.0291)
truncated Lévy (0,1, a,1,b)	-3282.9	0.8642 (0.1001) (0.1120)	1.7285 (0.0000) (0.0000)	0.9141 (0.0177) (0.0118)	0.0144 (0.0085) (0.0152)	0.0531 (0.0171) (0.0192)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	0.0000 (0.0000) (0.0000)	1.8087 (0.0610) (0.0741)	0.0715 (0.0629) (0.0700)	0.0300 (0.1327) (0.0956)

	HORIZON (DAYS)	VaR	S&P 500	NASDAQ	FTSE 100	HANG SENG	NIKKEI 225
EWMA-N	5	99%	2.6%	3.3%	2.9%	3.2%	0.8%
		95%	5.9%	7.6%	6.1%	8.1%	5.0%
	10	99%	1.8%	2.7%	1.1%	2.4%	0.7%
		95%	5.2%	6.6%	4.9%	8.6%	4.4%
[20	99%	0.9%	2.3%	1.3%	4.8%	2.0%
		95%	4.6%	7.2%	4.3%	8.9%	6.3%
AUGMENTED	5	99%	1.5%	1.2%	1.3%	1.3%	0.7%
GARCH- TRUNCATED		95%	5.2%	4.9%	5.3%	6.6%	4.0%
LÉVY	10	99%	1.2%	1.3%	0.8%	1.5%	0.7%
		95%	4.3%	5.5%	4.0%	6.0%	3.8%
	20	99%	0.7%	1.4%	0.9%	1.4%	1.3%
		95%	3.7%	4.6%	3.9%	6.1%	4.6%
Notes. This table 1% (5%) violatic for the second h	e presents viola ons on average alf of the perioc	tions of the for the 99% d 05/1992 u	expected multi-day (95%) level. The un ntil 03/2000 on daily	VaR (given an initial p derlying models have returns and a moving	osition of 100), where been estimated every window.	e the model assumes / day	

Table V: Out-of-Sample Analysis: Violations

